Author’s response to reviews

Title: A mixed methods quality improvement study to implement nurse practitioner roles and improve care for residents in long-term care facilities

Authors:

Kelley Kilpatrick (kelley.kilpatrick@mcgill.ca)

Éric Tchouaket (eric.tchouaket@uqo.ca)

Mira Jabbour (kkilpatrick.hmr@ssss.gouv.qc.ca)

Sylvie Hains (Sylvie.hains@hotmail.com)

Version: 2 Date: 08 Oct 2019

Author’s response to reviews:

Dear Professor Aronin, Dr Boyd, and Reviewer # 3

We would like to thank you for your comments and the opportunity to submit a revised manuscript # NURS-D-18-00264. R1. The elements that required more information or clarification have been reworked. All the questions have been addressed but not all the changes were made as requested. The changes are in yellow in the text. You will find enclosed a detailed list of changes that, we hope, will clarify the issues you have raised with the manuscript. The Table outlining the changes was uploaded as supplementary material.